Author's response to reviews

Title: Incidental findings on MRI scans of patients presenting with audiovestibular symptoms

Authors:

Vasileios Papanikolaou (vspap@hotmail.com)
Mohamed H Khan (mhkhan@hotmail.com)
Ivan J Keogh (ijkeogh@hse.ie)

Version: 3 Date: 20 May 2010

Author's response to reviews: see over
Dear Madam,

Once again we would like to thank the reviewers for their interest in our work and their additional comments. All suggestions have been made to the text, ie:

1. The abbreviation AVS has been removed from the text and the full text audiovestibular symptoms (reviewer no1)

2. The correction on page 10, line 16, please correct "n effective" to "an effective" has been made (reviewer no1 and no2)

3. The paragraph “So far MRI can not fully discover the pathological changes in the audiovestibular system, patients with negative results should be consulted in higher level hospital to get sophisticated inspection” has been added to the text (reviewer no2).

All corrections/changes have been highlighted with red in the text. Additionally the manuscript conforms to the manuscript formatting checklist

Sincerely yours,

The authors,